Author's response to reviews

Title: Parental educational level and cardiovascular disease risk factors in schoolchildren in large urban areas of Turkey: Directions for public health policy

Authors:

Bike Kocaoglu (kocaoglbb@boun.edu.tr)
George Moschonis (gmoschihua.gr)
Maria Dimitriou (ds2009@hua.gr)
Maria Kolotourou (ds20019@hua.gr)
Yasar Keskin (keskinyasar@yahoo.com)
Haydar Sur (haydars@yahoo.com)
Osman Hayran (hayran@yahoo.com)
Yannis Manios (manios@hua.gr)

Version: 6 Date: 14 January 2005

Author's response to reviews: see over
Dear BioMed Central Team,

We would like to express our appreciation to the editorial team for accepting our manuscript for publication to “BMC Public Health” journal. We believe that all the requested changes have been incorporated to the manuscript in the best possible way. However, do not hesitate to contact me for any reason, whatsoever.

Thank you very much for your support

Sincerely yours,

Yannis Manios